Author's response to reviews

Title: Influence of temporomandibular disorder on temporal and masseter muscles and association with occlusal contacts in adolescents: electromyographic study

Authors:

Leandro Lauriti (leandrolauriti@ig.com.br)
Lara J Motta (larajmotta@terra.com.br)
Camila HL Godoy (cami_godoy@hotmail.com)
Daniela A Biasotto-Gonzalez (dani_atm@uninove.br)
Fabiano Politti (fpolitti@uninove.br)
Raquel A Mesquita-Ferrari (raquel.mesquita@gmail.com)
Kristianne PS Fernandes (krisporta@gmail.com)
Sandra K Bussadori (sandra.skb@gmail.com)

Version: 2 Date: 10 June 2013

Dear revisors of BMC Musculoskeletal Disorders,

Please find enclosed the revised manuscript: “INFLUENCE OF TEMPOROMANDIBULAR DISORDER ON TEMPORAL AND MASSETER MUSCLES AND ASSOCIATION WITH OCCLUSAL CONTACTS IN ADOLESCENTS: ELECTROMYOGRAPHIC STUDY”, by Leandro Lauriti et al.

The changes were made as suggested and the comments are attached.

We hope that these changes have improved the manuscript.

Sincerely yours,

Camila H. L de Godoy on behalf of the authors.